Check for updates

Citation: Adebayo AS, Survayanshi M, Bhute S, Agunloye AM, Isokpehi RD, Anumudu CI, et al. (2017) The microbiome in urogenital schistosomiasis and induced bladder pathologies. PLoS Negl Trop Dis 11(8): e0005826. https://doi. org/10.1371/journal.pntd.0005826

Editor: Michael H. Hsieh, George Washington University, UNITED STATES

Received: January 31, 2017

Accepted: July 21, 2017

Published: August 9, 2017

Copyright: © 2017 Adebayo et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: ASA received travel support from The World Academy Academy of Sciences (TWAS)/ Department of Biotechnology (India) Postgraduate Fellowship in the course of the research. RDI acknowledges the award HRD-1435186 from the U.S. National Science Foundation. CIA acknowledges the World Health Organisation short term training grant B40394. The funders had no RESEARCH ARTICLE

The microbiome in urogenital schistosomiasis and induced bladder pathologies

Adewale S. Adebayo¹, Mangesh Survayanshi², Shrikanth Bhute², Atinuke M. Agunloye³, Raphael D. Isokpehi⁴, Chiaka I. Anumudu¹*, Yogesh S. Shouche²*

1 Cell Biology & Genetics Unit, Department of Zoology, University of Ibadan, Ibadan, Nigeria, 2 National Centre for Cell Science, Pune, India, 3 Department of Radiology, University of Ibadan, Nigeria, 4 Bethune-Cookman University, Daytona Beach, Florida, United States of America

* yogesh@nccs.res.in (YSS); chiaka.anumudu@mail.ui.edu.ng (CIA)

Abstract

Background

Human schistosomiasis is a highly prevalent neglected tropical disease (NTD) caused by *Schistosoma species*. Research on the molecular mechanisms influencing the outcomes of bladder infection by *Schistosoma haematobium* is urgently needed to develop new diagnostics, therapeutics and infection prevention strategies. The objective of the research study was to determine the microbiome features and changes in urine during urogenital schistosomiasis and induced bladder pathologies.

Methodology

Seventy participants from Eggua, southwestern Nigeria provided morning urine samples and were screened for urogenital schistosomiasis infection and bladder pathologies in a cross-sectional study. Highthroughput NGS sequencing was carried out, targeting the 16S V3 region. Filtered reads were processed and analyzed in a bioinformatics pipeline.

Principal findings

The study participants (36 males and 34 females, between ages 15 and 65) were categorized into four groups according to status of schistosomiasis infection and bladder pathology. Data analytics of the next-generation sequencing reads revealed that Proteobacteria and Firmicutes dominated and had influence on microbiome structure of both non-infected persons and persons with urogenital schistosomiasis. Furthermore, gender and age influenced taxa abundance independent of infection or bladder pathology. Several taxa distinguished urogenital schistosomiasis induced bladder pathologies from urogenital schistosomiasis infection alone and from healthy persons, including known immune-stimulatory taxa such as *Fusobacterium, Sphingobacterium* and *Enterococcus*. Some of these significant taxa, especially *Sphingobacterium* were projected as markers of infection, while several genera including potentially beneficial taxa such as *Trabulsiella* and *Weissella*, were markers of the noninfected. Finally, expected changes in protein functional categories were observed to relate to cellular maintenance and lipid metabolism. role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusion

The urinary microbiome is a factor to be considered in developing biomarkers, diagnostic tools, and new treatment for urogenital schistosomiasis and induced bladder pathologies.

Author summary

The human microbiome comprises bacteria (plus viruses, fungi and archeae) inhabiting different sites of the body. They do not specifically cause diseases, but their presence, absence or population influence body functions. We therefore examined such organisms found along the urinary tract, in persons living in a rural community in Nigeria who considered themselves healthy, were infected with the parasite *Schistosoma haematobium* or had developed bladder complications along with the parasite infection. We found that these groups shared a large portion of the microbiome, but there were microbial species unique to infected persons and those with bladder complication. Some of these were capable of inducing inflammation and could offer less protection to the host. We also predicted pathways that are affected by the difference in the microbiome.

Introduction

Human schistosomiasis is a devastating and highly prevalent neglected tropical disease (NTD) caused by Schistosoma species, a genus of parasitic flatworms with life cycle forms found in freshwater, freshwater snails and human organ systems [1]. Cercarial forms of the parasite are released from infected water snails and penetrate human skin. Eventually, the adult forms inhabit and produce eggs in the intestinal or bladder tissues. The cercariae, schistosomulum and adult stages are all capable of inducing host immune response [1-3]. The eggs of Schistosoma haematobium in the human bladder can cause a spectrum of urogenital clinical presentations including granulomatous inflammation, fibrosis, urinary tract infections and bladder cancer [2,3]. Seventy-eight countries and a quarter of a billion persons are at risk of schistosomiasis [4]. After considering the disease along with other NTDs, Nigeria was described as 'ground zero' of schistosomiasis due to the high endemicity in the country [5]. Few studies have reported on the immune response to urogenital schistosomiasis in Nigeria [6], with several studies in different parts of the country reporting prevalence rates between 15–57%[7, 8]. The occurrence and different forms of bladder tumors and bladder pathologies have been associated with urogenital schistosomiasis in Nigeria[9] and in other parts of Africa [2,10].

Research on molecular pathology influencing the outcomes of bladder infection by *Schisto-soma haematobium* is urgently needed to develop new diagnostics, therapeutics and infection prevention strategies[3].Earlier studies have suggested that the mechanism of formation of bladder tumours will be due to formation of nitrosamines, polyaromatic hydrocarbons, free radicals, and presence of microbes [11]. More recently, studies have highlighted the role of estrogen-related molecules from the parasite in disease progression [12], based on the discovery that they could be oxidized to form adducts [13], induce infertility in females [14], and could probably induce error-prone DNA repair [15]. Molecules related to estrogen were recently detected in urine samples of persons infected with schistosomiasis [16].

Our collaboration with communities in Nigeria at risk of schistosomiasis bladder-related pathologies presents opportunities to conduct basic and translational research projects[6,9,17].

We are interested in understanding the influences of microbial taxa on human health and disease [18] including the induction of bladder pathologies in urogenital schistosomiasis. Genomic sequencing technologies can determine the membership of, and functions performed by microbial communities (microbiome) in the urine [19–21]. The human microbiome is the community of microbes that is estimated to encode up to several million genes with the capacity to influence human health and disease[22]. The microbiome mediates the effectiveness of drugs, xenobiotics and vaccines[23], and influences disease or health status [24].

The microbiome along the urinary tract structure can have adverse or beneficial effects on human health [19–21]. Therefore, the goal of the research study reported in this article was to understand the microbiome features (taxa membership and functions of encoded microbial proteins) in the urine samples from healthy persons and persons infected with urogenital schistosomiasis and related pathologies.

The availability of volunteers with asymptomatic schistosomiasis in Eggua, in southwestern Nigeria allowed us to accomplish the research goal. Seventy volunteers were categorized into groups based on the presence of infection and induced bladder pathologies. The objectives of the study were to (1) determine the microbiome changes in urine samples during urogenital schistosomiasis and induced pathologies; and (2) identify functional biological processes that could be altered by such changes.

Methods

Overview of methods

The approach of the research study consisted of recruiting study participants; screening for schistosomiasis infection and bladder pathologies; microbiome sequencing; pre-processing of microbiome sequences; and data analytics of microbiome sequence data collection.

Ethics and study participants

Study participants were adults recruited from Eggua, in Ogun State, southwestern Nigeria (07° 01.592 N; 002° 55.083 E) in a cross-sectional study. Nearby communities have previous history of urogenital schistosomiasis and it was shown recently that the bladder pathologies occurrence in these localities is essentially driven by prevalence of urogenital schistosomiasis infection [7]. Ethical approval was obtained from University of Ibadan/University College Hospital Institutional Review Committee as well as the Ogun State Ministry of Health. Participants were duly informed about the study before sampling and informed consent obtained from all participants, with language translation as required. Exclusion criteria included recent use of antibiotics, painful bladder, urine discharge problems, common urinary tract infection. This was determined by interviews. While some of these criteria may indeed be due to urogenital schistosomiasis, it was necessary to prevent confounders and ambiguity. Samples suspected of urinary tract infection (positive nitrites, positive leukocyte esterase and \geq 5wbc/high power field) were removed from further analysis. Seventy samples meeting the exclusion criteria were eventually processed for high throughput sequencing.

Screening for infection and bladder pathologies

Medical history, routine diet and demographic factors were obtained via structured questionnaire, and participants provided midstream urine samples. All samples were collected in the morning hours and immediately anonymized upon collection. The presence of analytes in the urine samples was immediately determined with Urinalysis Reagent Strips (Rapid Labs, UK), and urine microscopy was done on a 10 ml aliquot to detect *S. haematobium* eggs after sedimentation; and the rest was immediately frozen until further use and transported under dry ice conditions when required. Egg shedding may be infrequent in chronic infection, hence if a sample was negative for *S. haematobium* eggs, it was subjected to PCR detection using published *Dra1* primers [25]. Briefly, 25ul PCR reaction mix containing 100ng isolated urine DNA was prepared. Cycling conditions were: initial denaturation at 95°C for 5mins, 30 cycles of 95°C for 30 secs, 55°C for 30 secs and 72°C for 1min, and final extension at 72°C for 10mins. Bladder scans were carried out with Titan UltraSystem (Sonosite, WA, USA) by a trained radiologist, and all images scored according to WHO recommendations [26] and also anonymized.

Sequencing and pre-processing

One ml of urine was pelleted and DNA was isolated using Qiagen Blood and Tissue kit (Qiagen, Hilden, Germany), with the modification of adding 20mg/ml of lysozyme at the lysis stage. Isolated DNA was quantified with NanoDrop spectrophotometer (Thermo Fisher Scientific, MA, USA), with quality assessed as 260/280 absorbance ratio >1.8. Library preparation and sequencing were done as previously described [24]. Briefly, quality control of sequencing library (size and quantity) was done on 2100 BioAnalyzer (Agilent Technologies, CA, USA) following manufacturer's protocol. Highthroughput sequencing targeting the V3 region of the 16S rRNA gene was carried out on IonTorrent PGM platform. Sequencing and barcoding were done using Ion PGM Sequencing 200 kit v2 and Ion Xpress Barcodes Adapters (Thermo Fisher Scientific, MA, USA) using manufacturer's protocol. Barcoded libraries purified with Agencourt AMPURE beads (Beckman Coulter, CA, USA), and equimolar amplicons were pooled prior to sequencing. V3 primers used were 343F- 5'TACGGRAGGCAGCAG3'and 533R- 5'TTACCGCGGCTGCTGGCAC 3'. Processing and quality filtering of sequence data was done using QIIME 1.9.1 [27]. Operational Taxonomic Units (OTUs) were defined based on 97% sequence similarity and taxonomic classifications were assigned using the Greengenes g_13_8 [28]. Where reference to the Greengenes database did not identify OTUs up to genus level, we used BLASTn to assign taxonomy based on 100% coverage and >98% identity, though this was possible for only two OTUs. Sequence data was examined for possible contamination from kits using the correlation between OTUs and amplicon concentration (purified individual library) [29, 30]. Sequencing was evaluated with Good's coverage and rarefaction plots.

Data analytics of microbiome sequence collection

Data was screened for missing values, normality and excessive outliers before applying statistical tools. After rarefaction to 2952 reads, biostatistical analyses were performed using STAMP [31], R packages Vegan [32] and Biom [33]. Public server at http://huttenhower.sph.harvard. edu/galaxy/ was used for LEfSe [34]. All tests were two-tailed; significant results were indicated by a p value < 0.05 and effect size estimated with Eta-squared. Mann-Whitney test (or non-parametric White's test [35] in STAMP) was used to test for significant differences between two groups. Comparison of multiple groups was done using non-parametric Kruskal-Wallis test and Tukey-Kramer post hoc test. Benjamini-Hochberg test/FDR was used for multitest correction. In consideration of the criticisms of normalization using rarefaction [36], which in this case would leave out 11 samples from analysis, we applied edgeR's RLE method [37] to normalize samples, for the fold change or differential abundance analysis of microbiome; thus including all 70 samples. Variance threshold was set at 1e-5 and FDR<0.05. Functional profiles were inferred with PICRUst [38].

Results

Overview of results

One of the key findings of this research was that microbial taxa membership and predicted protein function were uniquely discriminant for persons with urogenital schistosomiasis and those who were not infected. Data and project information are deposited in NCBI's Sequence Read Archives under accession SRP094688. A total of 1.4 million non-polyclonal, trimmed and filtered reads was obtained after quality control. Good's coverage was 98% and average conditional uncovered probability estimates was 0.05. The reads formed 2946 OTUs, excluding 1504 singletons which were removed before abundance and multivariate analyses. No OTUs were strongly and negatively correlated with purified amplicon concentration, but ten OTUs had moderate, negative correlation (rho = -0.3 to -0.4). This indicated that there was minimal level of probable contaminants from reagents. About 110 OTUs occurred in at least 55% of samples and only 1 OTU in all samples.

Seventy volunteers categorized into four groups according to status of infection and bladder pathology

The seventy samples analysed for urinary microbiome through NGS comprised 36 males and 34 females, between the ages of 15 and 65 years. Interviews revealed that the diet was essentially uniform, with a strong starch base. Bladder pathologies detected included abnormal thickness (>5mm of a full bladder), calcification, bladder mass, hyperplasia and irregular shape. We identified four major groups of participants based on the status of infection and bladder pathology (Table 1), comprising (a) the Advanced group, having infection and bladder pathologies, (b) Bladder pathology-only group, (c) infection-only (detected by microscopy or PCR), and (d) controls (no infection or pathology). For microbiome abundance analysis, we also grouped all samples into two—infected and non-infected.

Proteobacteria and Firmicutes dominate and influence urine microbiome structure

Differences in proportions of some phyla in the urine microbiome in the study population are presented in Figs 1 and 2A–2D. The mean relative abundance of two phyla, Proteobacteria

GROUPING	CATEGORIES	SAMPLE COUNT (PERCENTAGE)
AGE	ELDERLY (50years +)	25(36)
	MID-AGED (30–49 years)	30(43)
	YOUNG (<30years)	15(21)
INFECTION	NON-INFECTED	24(34)
	INFECTED	46(66)
DISEASE PROGRESSION	ADVANCED	22(31)
	PATHOLOGY-ONLY	10(14)
	INFECTION-ONLY	25(36)
	CONTROL	13(19)

Table 1. Participants in urine microbiome study from Eggua, southwestern Nigeria.

Advanced, urogenital schistosomiasis infection + abnormal bladder morphologies

Infection-only, urogenital schistosomiasis infection

Pathology-only, abnormal bladder morphologies

Controls, no infection or pathology.

https://doi.org/10.1371/journal.pntd.0005826.t001





(70%) and Firmicutes (26%) were the largest in the microbiome of the sample population, with Bacteroidetes, Actinobacteria, Fusobacteria, Cyanobacteria and others making up the rest (Fig 1, Fig 2A). The proportions of Proteobacteria differed among groups, highest in infection-only (65%) and pathology-only (80%) cases, and lowest in advanced cases (42%). The Proteobacteria proportion was higher in non-infected (73%) compared to infected (66%), and was unchanged in pathology (68%) compared to pathology-absent cases (69%). In all three comparisons, the opposite is true for Firmicutes. Virtually all of the Actinobacteria OTUs were found in controls and advanced cases. Having observed increased Firmicutes and decreased Proteobacteria in infected cases, the log of the ratio of the abundance of Firmicutes to Proteobacteria was calculated for each sample and plotted as a dysbiosis index. This index associated negatively with disease progression but the correlation was not statistically significant (rho = -0.29, p = 0.08) (Fig 2B). The most dominant genera were *Pseudomonas, Staphylococcus*,



Fig 2. (A) Mean relative abundance of various phyla in the urine microbiome of a study population in Eggua, southwestern Nigeria. (B) Bar plot of the Firmicutes/Proteobacteria log abundance ratio in different states of urogenital schistosomiasis. The correlation was not significant (p = 0.08). (C)Abundance of phyla Actinobacteria and Bacteroidetes with regard to gender in the urine microbiome of a study population in Eggua, southwestern Nigeria (p<0.05). (D) Proportions of filtered OTU counts in urogenital schistosomiasis and controls(p = 2.06E-16).

Acinetobacter, Enterococcus, unclassified Enterobacteriaceae and Facklamia (Fig 1). The first three, made up 60% of the whole community. Of the OTUs formed, the most abundant OTUs were those belonging to *Pseudomonas* and *Staphylococcus* (Fig 1). *Facklamia* was essentially found only in advanced cases, making up 12% in proportion. *Pseudomonas* proportions were increased by more than two-fold in infection-only and pathology-only cases (each at 47%) compared to advanced cases (19%) or controls (15%).

In terms of diversity of the samples, alpha diversity was estimated using the inverse Simpson's index (1/D); this measure takes both OTU richness and relative abundance into account and the results are presented in Fig 3A–3C. We found a non-significant reduction in mean diversity in infected cases compared to non-infected, in pathology cases compared to nopathology cases, and in advanced cases compared to others (p = 0.1) (Fig 3A and 3B). There was higher number of OTUs detected in infected, pathology and advanced cases, respectively in each case. This indicated that advanced cases had higher number of OTUs but reduced diversity index compared to the other three groups. The same was observed for infected cases compared to non-infected cases.

Beta diversity was assessed using Bray-Curtis metrics to estimate dissimilarities based on OTU level, rarefied data. Similar patterns were observed considering infection and pathology. The first two principal coordinates explained 36% of the variation among them, and a plot of the first two axes revealed little level of axes separation among the sample groups (Fig 4A). Of the ten most dominant OTUs, 4 of them, all assigned to the genus *Pseudomonas*, were prevalent (and in opposition to other prevalent OTUs) along the first axis (Fig 4A). They clustered close to several infected or pathology samples. Four other OTUs, all assigned to *Staphylococcus*





were in contrast, driving the axes in the opposite direction, and clustering close to some control samples (Fig 4A).

Gender and age could influence taxa abundance independent of infection or bladder pathology

Mean diversity of organisms differed among age groups (Fig 3C), and the middle-aged group contained 33% more OTUs than the young or elderly age groups. Since presence of infection or pathology could be expected to affect taxa differences in different ways among age groups and prevalence rates were different among the age groups, we tested taxa differences among age groups using only controls (n = 13), There were no significant taxon differences at all levels tested, which was probably due to low sample size for comparison. We therefore used partial canonical correspondence analysis to explain taxa differences with age of participants. This will involve all samples and statistically exclude the influence of being infected or having pathology. To achieve this, the age groups were arranged into smaller ranges i.e. age 10–20 (teens), 20–30, up to 60 and above. This was to increase the number of age groups to six, which was considered better in this type of analysis than having three groups. Canonical



Fig 4. (A) Principal Coordinates biplot of beta diversity in different states of urogenital schistosomiasis and controls. Circle and small typeset represent Greengenes ID of dominant OTUs. These dominant OTUs were mainly assigned to *Pseudomonas* and *Staphylococcus*, but had little influence on disease status. (B) Partial canonical correspondence analysis of age groups without the influence of urogenital schistosomiasis infection. Blue fonts are age categories, red typeface are Greengenes ID of the most influential OTU. Clustering together indicates no special correlation with age group. OTUs that corresponded closely with participants aged sixty years were mainly assigned to *Acinetobacter*.

correspondence analysis was then mapped using the age ranges as constraining variables and infection as conditioning variable. This was done in order to parse the effect of being infected or not on each taxon. Only OTUs present in at least 40% of samples were utilized. Four OTUs, all assigned to genus *Acinetobacter* appear sensitive to age 60 and above, those assigned to *Citrobacter*, *Enterococcus*, and Enterobacteriaceae were sensitive to age 30–40 (Fig 4B). Indeed, age group 30–40 could be influenced by many OTUs. Similar results were obtained with pathology as the conditioning variable.

There was a difference between sexes in the microbial community at the OTU and genus levels. There was more abundance of Actinobacteria and Bacteroidetes phyla in females than males (Fig 2C). More heterogeneity was observed in females, with 40% more OTUs present compared to males. There was a non-significant reduction in diversity in female samples. Using only control samples, unclassified Enterobacteriaceae and unclassified Pseudomonadaceae were significantly abundant in females (p = 0.035, p = 0.021, respectively).

Several taxa distinguish urogenital schistosomiasis infected cases from non-infected cases, as well as advanced cases from infection alone

There was a significant difference in the microbial community between infected and noninfected, at all levels tested, genus level (p = 6.55E-10) and OTU levels (p = 2.06E-16) (Fig 2D). Several features (taxa) could distinguish persons infected with urogenital schistosomiasis and non-infected persons. Differences in the abundance of microbiome taxa in the study groups are depicted in Fig 5A–5D, S1 Fig and S2 Fig. At phylum level, the differences in the dominant phyla, Proteobacteria and Firmicutes, was not statistically significant between the infected and the non-infected. Phylum Fusobacteria, was the only significant phylum (FDR = 0.006), to



Fig 5. Specific urinary microbes are abundant in different states of urogenital schistosomiasis infection. (A) Differentially abundant microbiome families in urogenital schistosomiasis (infected) and controls (non-infected), (B) Differentially abundant microbiome genera in urogenital schistosomiasis (infected) and controls (non-infected), (C) Differential microbiome genera between urogenital schistosomiasis (infection-only) and urogenital schistosomiasis induced bladder pathology (advanced), (D) Differentially abundant microbiome genera in non-infected bladder pathology (pathology-only) and urogenital schistosomiasis induced bladder pathology (advanced) (FDR<0.05). Differential abundance was measured with LogFC, the log2 of the number of times the OTUs belonging to a genus (or family) are more numerous in one group relative to the other. Circle on a vertical line represents a bacterial genus or family colored by their phylum and the genus or family is named at the end of the line. More than one circle on a vertical line represents species of the same genus. The genus is labeled on the x-axis. UC represents a genus whose identity could not be completely confirmed, but with known family or order. Abundant microbes in each of the two groups are presented on either side of the middle zero line.

distinguish infected samples from non-infected samples. At family level, twenty-two families showed differential abundance between infected and non-infected. The abundance of *Veillo-nellaceae*, Fusobacteriaceae, Lactobacillaceae and Enterococcaceae were among those significantly higher in infected cases, while Oxalobacteraceae, Enterobacteriaceae, Staphylococcaceae among others were significantly abundant for non-infected samples (FDR<0.05) (Fig 5A). At the genus level, bacterial genera most significantly abundant in infected samples included *Facklamia, Fusobacterium, Veillonella, Bacteroides, Lactobacillus* and *Enterococcus*. On other hand, *Bacillus, Staphylococcus, Janthinobacterium, Edwardsiella*, unclassified Bacillaceae, unclassified Enterobacteraceae, *Trabulsiella, Xenorhabdus, Collimonas* and *Weissella* were significantly associated with non-infected samples (FDR<0.05) (Fig 5B). Different *Acinetobacter*



Fig 6. Urogenital schistosomiasis and induced bladder pathology result in specific microbiome changes in the urinary tract. Differentially abundant microbiome (A) between urogenital schistosomiasis induced bladder pathology (advanced) and healthy controls, and (B) between urogenital schistosomiasis (infection-only) and healthy controls (FDR<0.05). Differential abundance was measured with LogFC, the log2 of the number of times the OTUs belonging to a genus (or family) are more numerous in one group relative to the other. Circle on a vertical line represents a bacterial genus or family colored by their phylum and the genus or family is named at the end of the line. More than one circle on a vertical line represents species of the same genus. The genus is labeled on the x-axis. UC represents a genus whose identity could not be completely confirmed, but with known family or order. Abundant microbes in each of the two groups are presented on either side of the middle zero line.

OTUs were associated with both infected and non-infected samples (Fig 5B). Only some of these genera, especially *Fusobacterium* and *Janthinobacterium*, were found to be highly abundant in infected or non-infected samples, respectively, if rarefaction was applied before analysis (S2A Fig).

Comparing controls to other groups, unclassified Clostridiales (p = 0.047, η^2 = 0.11) was highest in controls, and lowest in infection-only and pathology-only cases (S1A Fig). Enterobacteriaceae (0.24 \geq $\eta^2 \leq$ 0.29, 0.022 \geq p \leq 0.039) was more abundant in controls (S1B Fig), and Pseudomonadaceae (0.14 \geq $\eta^2 \leq$ 0.31, 0.025 \geq p \leq 0.048) far more abundant in infection-only cases and pathology-only cases (S1C Fig). The correlation of *Lactobacillus* abundance and healthy status was negative (rho = -0.34, p = 0.039) (S1D Fig).

Five genera—*Facklamia*, *Veillonella*, *Fusobacterium*, *Bacteroides*, and *Aerococcus*—were the most differentially abundant in advanced cases, when compared with infection-only cases, pathology-only or control cases (Fig 5C and 5D, Fig 6A). *Aerococcus*, *Acinetobacter* and *Staphylococcus* were most changed in infection-only cases (Fig 6B, S2B Fig), though other OTUs of *Acinetobacter* and *Staphylococcus* were highly associated with control cases. The least changes in taxa between groups were in the pathology-only: control pair and the pathology-only: infection-only pair.

Linear discriminant analysis prediction highlighted fewer taxa as biomarker

Microbiome sequence data was subjected (at OTU and genus levels) to discriminant analysis using LEfSe to identify possible biomarkers. Comparing infected and non-infected cases, all taxa that were significantly associated with each group from earlier significance analysis were also identified at the OTU level. At genus levels, only two of the significant taxa, *Sphingobacter-ium* and *Aerococcus* were projected as markers of infection, while several genera including *Tra-bulsiella* and *Weissella*, associated with non-infected (Fig 7).



Fig 7. Predicted microbial biomarkers in urogenital schistosomiasis cases (infected) and controls (non-infected).

Predicted changes in functional categories relate to cellular maintenance and lipid metabolism

Using our sequence abundance data, we obtained imputed metagenomes and the associated KEGG Orthology pathways present in the microbiome. Nearest Sequenced Taxon Index (NSTI) summary ranged from 0.021 to 0.053 (mean = 0.039). This indicated that several closely related genomes were available for inference purposes. A total of 328 functional categories from 6908 KOs were identified.

Of the features differentiating infected samples, several were significant (Fig 8) between infected and non-infected cases, though the effect sizes were rather small. This may be expected given that large portions of microbiome were shared. Considering all four groups together, 7 features which were statistically different—including lipid metabolism (reduced in advanced cases) and alpha-linoleic acid metabolism (reduced in advanced and infection-only cases) ($0.1 < \eta^2 > 0.143$). In pairwise comparison with other groups, the more abundant features with the best effect sizes in advanced cases were pathways involving DNA maintenance including mismatch repair ($\eta^2 = 0.28$), DNA recombination/repair, and translation factors (S3 Fig).

Discussion

The overall microbiome structure from our data is, in part, similar to those of earlier studies [19, 20, 39, 40]. These studies also showed that the two phyla, Proteobacteria and Firmicutes, which form the core of the microbiome in the present study, occur in human urinary microbiome. But, in contrast to the aforementioned studies, low proportions of Firmicutes, very high proportion of Proteobacteria, and very low proportions of Actinobacteria /Bacteroidetes, are the features which clearly differentiate the microbiome data of the present study. The

95% confidence intervals



Fig 8. Predicted biochemical changes in the KEGG Ortholog database due to microbiome differences between urogenital schistosomiasis cases (infected) and controls (non-infected).

https://doi.org/10.1371/journal.pntd.0005826.g008

greater abundance of Actinobacteria and Bacteroidetes in females as reported previously [41], was also observed in the current study. Despite the low occurrence of these two phyla, the number of their OTUs in the sequence data of the present study was at least three times higher in females than males in this study (Fig 2C). In addition, the dominance of the Proteobacteria and Firmicutes was found in all samples (Fig 1, Fig 2A), unlike in the previous reports mentioned [19, 20, 39] where only few samples had such dominance. The study location is probably of importance in these contrasts; the present study involves a wholly rural and African population, while all these aforementioned reports have been European, Australian or North American. This raises the question of whether a large reduction in Firmicutes and consequent increase in Proteobacteria could be characteristic of a rural African urinary microbiome. Also, the higher proportion of Proteobacteria in non-infected cases, and its lower proportion in advanced cases compared to infection-only or pathology-only, indicates that the dominance of Proteobacteria in our population is not linked with disease conditions.

Another observation of the microbiome structure is the relatively high number of singleton OTU (1504 out of 4451). This may be indicative of considerable level of individual uniqueness in urinary microbiome composition, though it is also possible that a greater depth of sequencing will reduce the number of singletons. Such an indication of individual uniqueness is buttressed by the nature of the principal coordinates plot (Fig 4A). The spread of the coordinates indicates that there was substantial within-group variation, which therefore affected between-group axis separation. With this, our data shows there is individual uniqueness in the urinary tract microbiome composition.

In a study [42], two of the most dominant genera from our data, *Pseudomonas* and *Staphy-lococcus*, were isolated in only 2% and 18%, respectively, of 920 school children infected with urogenital schistosomiasis in Ibadan, southwestern Nigeria [42]. The study, which was entirely culture based, also highlighted that common urinary tract infection dominated by *E.coli* and *Klebsiella* may associate with urogenital schistosomiasis infection. In the current study, we systematically removed samples suspected of UTI, in order to focus on urogenital schistosomiasis alone and to reduce the possibility of analyzing co-infected samples. In addition, we used a high throughput DNA sequencing approach. Thus, our study was different and shows that *Pseudomonas* and *Staphylococcus* species are widespread in the population of interest and not simply because of infection.

There was a trend of reduction in the mean diversity of the microbiome in infection and advanced cases, and previous studies have reported such a reduction in diversity in disease conditions; such as in cystitis [43] and in inflammatory bowel disease [44]. In the current study, the reduction in diversity is most likely due to presence of *S haematobium* infection because there is increased diversity in pathology-only cases (n = 10). Hence the reduction in diversity in advanced cases is not likely to be because of the contribution of pathologies, but rather of *S haematobium* infection. The observation of a reduction in diversity, coupled with increased number of OTUs, may indicate that almost all species in the urine microbiome were covered by the sequence data and an increase in total sequence collection may not substantially increase microbiome diversity. For age groups and gender, higher number of OTUs was observed in middle aged and in females, indicating greater heterogeneity in females and the middle-aged. This heterogeneity is not due to infection or pathology because in terms of gender, males had more infection and pathology than females, and the middle-aged group had similar infection or pathology rates with other groups.

Associating specific taxon with disease state is an important goal in microbiome studies. For this purpose, the results of the present study were mainly examined at the family and genus levels. A substantial number of abundant genera are pathobionts, including–*Enterococcus, Bacteroides, Facklamia* (increased in infected) and *Enterobacter, Chryseobacterium, Edwardsiella*, etc (increased in controls). However, based on literature and our functional analysis, several of the dominant genera are biologically relevant within the environment. Some of them have been previously found in association with a non-healthy status. *Enterococcus* was associated with neuropathic bladder patients who are at risk of asymptomatic bacteriuria [45]. A previous study [20] identified *Pseudomonas* (in males only), *Staphylococcus* (in both genders), *Enterococcus* and *Facklamia* (females only) in urine samples of study subjects who were 'partially healthy' i.e. without any urinary tract disease or symptoms, but having other nonspecified ailments. Also, their individual attributes as a result of specialty genes, proteins or pathways could be of importance in the urinary microbiome, as discussed subsequently.

Clearly, the dominant genus, *Pseudomonas*, in the microbial community did not significantly differ among groups and it may then be suspected that the genus had no influence on urogenital schistosomiasis or induced bladder pathologies, especially, since the *Pseudomonas* genus is a known opportunistic pathogen in common urinary tract infections [46]. However, a point of interest is the ability of some strains, including *P. fragi and P. putida* (along with very few Proteobacteria strains), to utilize extraneous steroids such as estrogen, due to the presence of specialized genes, in particular, tesD [47]. This is of interest because a recent report indicated the availability of catechol estrogens, steroid-like molecules, in the urinary tract during urogenital schistosomiasis infection [16]. In the present study, apart from being a dominant genus, the proportions of *Pseudomonas* and *P. fragi* were higher in infected cases. Some other OTUs assigned to the same genus were also significantly higher (0.00799> p<0.042). We hypothesize that the presence and proportion of some strains of *Pseudomonas* may be linked to the availability of greater amounts of steroid-related lipids or other lipid metabolites in the urine of infected persons and those with induced pathologies. Such a hypothesis requires further investigation involving gene expression studies, or reporter assays and strain identification for evaluation and confirmation.

Still on the dominant genera, the presence of different sets of OTUs of *Acinetobacter* or *Staphylococcus* in infected or advanced cases and in non-infected controls indicate a species-specific abundance in the two conditions. It is difficult to delve deeper into such observation, because longer length sequences than those used in this study are needed to confidently identify species-specific difference in the two opposing conditions. Nevertheless, it is known that some *Acinetobacter* species, for instance, *A. johnsonii* can destroy worms while others are pathobionts [48].

We found that genera with roles in the inflammatory process of human and other mammalian hosts associated strongly with and were differentially abundant in infected and advanced cases. The genus Fusobacterium (up to family, order and phylum level) consistently associated with advanced cases but not infection-only cases. Fusobacterium possesses potent lipopolysaccharides, are known to recruit immune cells [49] and are ubiquitous in colon cancer biopsies [50]. The genus is also known to serve as anchor for biofilm formation [49]. Sphingobacterium, identified as a marker in urogenital schistosomiasis, is known to possess unusual, non-mammalian sphingolipids containing ceramides. Such compounds are immunogenic and can activate macrophages, thereby contributing to the inflammatory process [51]. Human IgE antibody regulation could be affected in allergy or atopy, by Enterococcus [52,53]. We found elevated proportions of *Enterococcus* in infected persons, and increased IgE levels were associated with urogenital schistosomiasis [54]. It raises the question as to whether Enterococcus prevents excessive buildup of IgE in urogenital schistosomiasis infection. Thus, a future research objective would be to determine the influence of *Enterococcus* species on human IgE antibody production in urogenital schistosomiasis. To add to this, Bacteroides numbers are sometimes correlated with inflammation, and its lipopolysaccharides are potent immune-stimulators, yet its polysaccharide A is also capable of repressing pro-inflammatory cytokine [55, 56]. The four genera discussed above were differentially abundant in advanced cases, implying that their abundance is infection-related. All these suggest a possible role for these taxa in the maintenance or inducement of bladder pathologies in schistosomiasis, a disease in which inflammation is a known driver of complications. Such a role will complement other possible inducers of inflammation such as egg oviposition by the parasite. In summary, we have identified few microbial taxa that could regulate the maintenance or initiation of bladder pathologies in urogenital schistosomiasis.

A translational application of microbes is in the treatment of bladder tumours. BCG vaccine (attenuated *Mycobacterium tuberculosis*) when injected into the bladder reduced recurrence and progression in half of bladder cancer patients under study [57]. Administration of *L. casei* or *L. rhamnosus* in rodent models decreased progression of bladder tumour cells [58]. Both studies [57,58] were concerned with bladder tumours that were not due to schistosomiasis. We hypothesize that in *S. haematobium* induced bladder cancer, strategies to deplete these bacteria taxa combined with BCG vaccine might improve efficacy of treatment at least in some individuals.

Another taxon consistently associated with infection state in our data is the genus *Lactoba-cillus* (up to family and order level). High *Lactobacillus* proportion has been identified in both urine and vaginal microbiome studies, though it tends to be more prevalent in females [41]. This was far from the case in the present study. Rather, our data suggests that the microbiome in our study population belong to the 'diverse' or low *Lactobacillus* group urinary microbiome. As in the current study in which *Lactobacillus* is associated with infected samples, another

study [19] also reported increased abundance of this genus in interstitial cystitis. Our study supports the emergent theory that rather than view the whole genus as symbiotic, certain species are more frequent in disease state, for instance, *L. gasseri* [39], and could probably be considered pathobiont. A proposed mechanism to explain these observations is that *Lactobacillus* tolerance to common anti-bacterial compounds allows it to multiply easily and invade tissues to cause inflammatory changes [59].

Some of the significant genera found in the present study have been previously discovered to be associated with healthy or disease state in urine microbiome studies. *Dialister* and *Gemella* were previously associated with STI [41]. *Dialister* and *Facklamia* were more frequent in incontinence, and Enterobacteriaceae and *Staphylococcus* were more frequent in healthy controls [39]. *Enterococcus* associated with cystitis [43], but *Dialister* and *Staphylococcus* have been found in healthy female urine [19]. Our data agree with these earlier studies in the sense that urine microbial genera found to be frequent in a disease state by earlier authors—*Facklamia* and *Enterococcus* were also associated with urogenital schistosomiasis. The aforementioned studies indicated that *Dialister* and *Gemella* may occur frequently in either disease or healthy state, but in the current study they were more frequent in infection. Other genera previously described in urine samples such as *Corynebacterium* and *Atopobium* were all observed in our data, but did not associate with disease or healthy state. The current study has helped to confirm earlier studies on the frequency of some urinary microbial genera in disease state.

Some of the significant taxa in the microbial community abundant in control samples (whether non-infected, non-pathology or both) are a mixture of pathobionts as well as beneficial and protective species. Enterobacteriaceae have abilities in tolerance of inflammation and redox stress; *Collimonas* (anti-fungal); *Weissella* (probiotic properties, anti-fungal, antibiotic, stress resistance); *Janthinobacterium* (antibiotic, anti-fungal, anti-tumour, stress tolerance); *Trabulsiella* (type IV secretory system to prevent competition) and unclassified Lactobacillales (lactic acid bacteria group)[60–66]. Thus, even in the absence of high proportions of *Lactobacillus sp*, there could be beneficial microbes in the urinary tract. Whether this is unique to our study population or not requires further investigation. In summary, it may be said that bacterial genera with high tolerance, antibiotic and anti-fungal properties (including lactic acid bacteria), were significant residents in control samples. As the results show, some other beneficial species may occur even in advanced cases and such may help to reduce the extent of damage caused.

Microbial-related KEGG Ontology (KO) pathways obtained from the analysis of imputed metagenomes are of biological importance. The biochemical pathways are essentially predicted from our sequence data and the proteins involved may not be translated, yet the links in predictions merit an examination. It appears that pathways to enhance proliferation and rapid multiplication of cells are increased in the infected samples compared to non-infected, and in advanced cases compared to other groups (S3A Fig, Fig 8). Hence, repair, transcriptional or translational factors and nucleotide metabolism are significantly higher in infected cases, and even higher in advanced cases (S3A Fig). Multiplication of microbes is sure to induce action from the immune system in the human host and this may aggravate infection. Hence, it is of concern that this proliferation ability increases with disease progression (i.e. from controls to infected to advanced), as our results suggest.

In addition, our results suggest decreased biosynthesis of some lipids in infected compared to controls, and in advanced cases compared to other groups. Hence, pathways significantly more abundant in non–infected cases are those that involve synthesis of several lipids involving quinones, terpenoids and steroids. An explanation for the decreased representation of lipid metabolism, synthesis, or modification pathways in infected and advanced cases could be the abundance of such molecules in the environment (since it has been reported that *S.haematobium*

produces steroid-like lipids into the urinary tract). The question arises especially because it is known that microbes may switch on/off regulatory mechanisms in the presence of needed materials. Further research is clearly needed to explore this relationship.

Some of the significant biochemical pathways from this study have also been highlighted in earlier studies. In a review of empirical studies, Borningen et al. [67] highlighted the involvement of riboflavin metabolism in IBD (Inflammatory Bowel Disease), nucleotide and lipid metabolism in type1 diabetes. In the current study, our analysis suggests reduced riboflavin and lipid metabolism, and like IBD, urogenital schistosomiasis is also inflammation driven, leading to induced pathologies. Hence, while IBD is not a parasitic disease, it appears part of the inflammatory process involving microbes may be shared with urogenital schistosomiasis. Furthermore, there is a noticeable link among the microbial pathways whose gene abundances are reduced in infected samples. Alpha-linolenic acid is known to be produced by lactic acid bacteria, it can inhibit shikimate kinase, a precursor to the formation of chorismate (KEGG C00251) in the shikimate pathway. 4-Hydroxybenzenoate (KEGG C00156) from chorismate is used by bacteria in the biosynthesis of ubiquinone and other-terpenoid-quinone, a process more abundant in non-infected cases. Also, isochorismate (KEGG C00885) from chorismate is essential for biosynthesis of siderophore group non-ribosomal peptides, another process more abundant in non-infected cases. A product of 4-Hydroxybenzenoate, hydroquinone, in its modified form, is a byproduct in riboflavin synthesis. This linkage is probably important and could depend heavily on the shikimate pathway and chorismate.

It is known that several downstream products of chorismate are useful as antimicrobials [68], and that the ubiquinones, terpenoid-quinones and riboflavin play important roles as coenzymes in oxidative phosphorylation and electron-carrier system. Also, the siderophore group non-ribosomal peptides include several catechol-based molecules for import of iron chelation, importation of xenobiotics into the cell among other functions [69, 70]. This is in addition to the fact that some transporters are also significantly reduced. Thus, the abundance of microbial biological processes is altered in the course of urogenital schistosomiasis and bladder pathologies. An explanation for the altered abundance is that dysbiosis brought about by loss or reduction of protective, beneficial bacteria creates a more stressful and less controlled microbiome during urogenital schistosomiasis infection.

The use of midstream urine as a representative of the bladder surface from which it is discharged, is not without disagreement in literature [43, 71]. The arguments about the methods are fuelled by the possibility of contamination especially in samples from females or the elderly; hence contaminant removal is crucial. In this study, none of the elderly participants had any impairment and with proper instructions to participants, we believe that contamination was minimal. Hence, the samples analyzed in this study comprise the bladder surface microbiome. Also, in this study, virtually all genera reported from studies using transurethral catheterization to sample bladder microbiome [39] were observed. In addition, given that other forms of bladder pathology exist, a vast majority of the pathology cases recorded here are induced by urogenital schistosomiasis infection. In our study locality, bladder pathology was extensively associated with urogenital schistosomiasis [9]. It is also possible the pathology cases with no infection (determined by urine microscopy only), could be due to a chronic urogenital schistosomiasis infection in which egg shedding reduces progressively with occurrence of pathologies.

To summarize, while the current study substantially reports differences in the urinary tract microbiome between persons infected with urogenital schistosomiasis and healthy persons, and between persons having bladder pathologies due to urogenital schistosomiasis and persons with urogenital schistosomiasis infection alone, there are a few possible limitations. It remains to be confirmed if there is long-term stability in the microbiome observed, although

the infected cases in the present study can be classified as chronic infection. Another limitation may be the small sample size.

Conclusion

In this study, we examined the urinary microbiome in urinary or urogenital schistosomiasis and its induced bladder pathologies. We found that the urinary tract microbiome of the whole study population itself differs from earlier studies elsewhere. We also detected several beneficial and stress tolerant taxa in control cases, and immune-stimulatory taxa in urogenital schistosomiasis infected and urogenital schistosomiasis associated bladder pathology cases. In the microenvironment of the bladder and urinary tract, these changes probably cause lower chorismate production, synthesis of some lipids and promote self-proliferation. The urinary microbiome is a factor to be considered in developing biomarkers, diagnostic tools, and new treatment for urogenital schistosomiasis and induced bladder pathologies.

Supporting information

S1 Fig. Proportions of selected urine microbiome taxa in different states of urogenital schistosomiasis and controls. Differences in proportions were significant (p<0.05) (A) Clostridiales (B) Pseudomonadaceae, (C) Enterobacteriaceae, (D) Lactobacillus. (TIF)

S2 Fig. (A) Changes in the urine microbiome of urogenital schistosomiasis (infected) compared to controls (non-infected) with rarefaction of samples prior to analysis. (B) Differential abundance in the urine microbiome of persons with urogenital schistosomiasis infection without bladder pathology (infection-only) and pathology without schistosomiasis infection (pathology-only) (FDR<0.05).Differential abundance was measured with LogFC, the log2 of the number of times the sequences belonging to a genus (or family) are more numerous in one group relative to the other. Circle on a vertical line represents a bacterial genus or family colored by their phylum and the genus or family is named at the end of the line. More than one circle on a vertical line represent species of the same genus. The genus is labeled on the x-axis. UC represents a genus whose identity could not be completely confirmed, but with known family or order. logFC is the logarithm of the fold change between two groups. Abundant microbes in each of the two groups are presented on either side of the middle zero line. (TIF)

S3 Fig. Predicted biochemical changes due to microbiome differences in (A) urogenital schistosomiasis induced bladder pathology (advanced) and urogenital schistosomiasis infection alone (infection), (B) urogenital schistosomiasis induced bladder pathology (advanced) and pathology without infection (b.path), (C) urogenital schistosomiasis induced bladder pathology (advanced) and healthy controls (control). (TIF)

Author Contributions

Conceptualization: Adewale S. Adebayo, Atinuke M. Agunloye, Raphael D. Isokpehi, Chiaka I. Anumudu.

Formal analysis: Adewale S. Adebayo, Raphael D. Isokpehi, Chiaka I. Anumudu.

Investigation: Adewale S. Adebayo, Shrikanth Bhute, Atinuke M. Agunloye, Chiaka I. Anumudu.

Methodology: Adewale S. Adebayo, Mangesh Survayanshi, Shrikanth Bhute, Yogesh S. Shouche.

Project administration: Chiaka I. Anumudu.

Visualization: Adewale S. Adebayo, Raphael D. Isokpehi.

- Writing original draft: Adewale S. Adebayo, Chiaka I. Anumudu.
- Writing review & editing: Adewale S. Adebayo, Shrikanth Bhute, Atinuke M. Agunloye, Raphael D. Isokpehi, Chiaka I. Anumudu, Yogesh S. Shouche.

References

- Colley DG, Bustinduy AL, Secor WE, King CH. Human schistosomiasis. The Lancet 2014; 383: 2253– 2264.
- Rambau PF, Chalya PL, Jackson K. Schistosomiasis and urinary bladder cancer in North Western Tanzania: a retrospective review of 185 patients. Infectious Agents and Cancer 2013; 8:19. <u>https://doi.org/ 10.1186/1750-9378-8-19 PMID: 23705833</u>
- Ray D, Nelson TA, Fu CL, Patel S, Gong DN, Odegaard JI, et al. Transcriptional profiling of the bladder in urogenital schistosomiasis reveals pathways of inflammatory fibrosis and urothelial compromise. PLoS Neglected Tropical Diseases 2012; 6: e1912. https://doi.org/10.1371/journal.pntd.0001912 PMID: 23209855
- World Health Organization (WHO). Schistosomiasis Fact sheet. 2014. Available from www.who.int/ mediacentre/factsheets/fs115/en/
- Hotez PJ, Asojo OA, Adesina AM. Nigeria: "Ground Zero" for the High Prevalence Neglected Tropical Diseases. PLoS Neglected Tropical Diseases 2012; 6(7): e1600. https://doi.org/10.1371/journal.pntd. 0001600 PMID: 22860138
- 6. Anumudu CI, Alabi O, Oniya MO. Schistosome specific antibodies in individuals co-infected with malaria in Southwest Nigeria. Nigerian Journal of Parasitology 2012; 33(2): 133–139.
- Ladan MU, Abubakar U, Abdullai K, Bunza MDA, Ladan MJ, Adamu T, et al. Gender and age-specific prevalence of urogenital schistosomiasis in selected villages near a dam site in Gusau Local Government Area, Zamfara state. Nigerian Journal of Parasitology 2011; 32(1): 55–60.
- Ekejindu IM, Ekejindu GO, Andy A. Schistosoma haematobium infection and nutritional status of residents in Ezi-anam, a riverine area of Anambra state, south-eastern Nigeria. Nigerian Journal of Parasitology 2002; 30(2): 96–101.
- Onile OS, Awobode HO, Oladele VS, Agunloye AM, Anumudu CI, et al. Detection of urinary tract pathology in some *Schistosoma haematobium* infected Nigerian adults. Journal of Tropical Medicine 2016 Jul 16. https://doi.org/10.1155/2016/5405207 PMID: 27635146
- 10. Zaghloul MS, Gouda I. Bladder cancer and Schistosomiasis: Is there a difference for the association? InTechOpen 2012 Feb 1. https://doi.org/10.5772/26819
- 11. Mostafa MH, Sheweita SA, O'Connor PJ. Relationship between schistosomiasis and bladder cancer. ClinMicrobiol Rev. 1999; 12: 97–111.
- Botelho MC, Vale N, Gouveia MJ, Rinaldi G, Santos J, Santos LL, et al. Tumour-like phenotypes in urothelial cells after exposure to antigens from eggs of *Schistosoma haematobium*: An oestrogen–DNA adducts mediated pathway? International Journal for Parasitology 2013; 43: 17–26. https://doi.org/10. 1016/j.ijpara.2012.10.023 PMID: 23260770
- Gaikwad NW. Metabolomic profiling unravels DNA adducts in human breast that are formed from peroxidase mediated activation of estrogens to quinonemethides. PLoS One 2013; 8(6): e65826. <u>https://doi.org/10.1371/journal.pone.0065826</u> PMID: 23762435
- Santos J, Gouveia MJ, Vale N, Delgado ML, Goncalves A, daSilva JM, et al. Urinary estrogen metabolites and self-reported infertility in women infected with *Schistosoma haematobium*. PLoS ONE 2014; 9 (5): e96774. https://doi.org/10.1371/journal.pone.0096774 PMID: 24848950
- Cavalieri EL, Rogan EG. Unbalanced metabolism of endogenous estrogens in the etiology and prevention of human cancer. J Steroid BiochemMolBiol 2011; 125: 169–180.
- Botelho MC, Alves H, Richter J. Estrogen catechols detection as biomarkers in schistosomiasis induced cancer and infertility. Lett Drug Des Discov 2017; 14(2): 135–138. PMID: 28018158

- Oniya M, Omosun Y, Anumudu C, Nwuba R, Odaibo A (2011) IgG Enzyme-Linked Immunosorbent Assay (ELISA) for immunodiagnosis of *Schistosoma haematobium* infected subjects living in an endemic Nigerian village. Scientific Research and Essays 6: 1650–1655.
- Isokpehi RD, Udensi UK, Simmons SS, Hollman AL, Cain AE, Olofinsae SA, et al. Evaluative profiling of arsenic sensing and regulatory systems in the human microbiome project genomes. Microbiology Insights 2014; 7: 25. https://doi.org/10.4137/MBI.S18076 PMID: 25452698
- Siddiqui H, Nederbragt AJ, Lagesen K, Jeansson SL, Jakobsen KS. Assessing diversity of the female urine microbiota by high throughput sequencing of 16S rDNA amplicons. BMC Microbiol. 2011; 11: 244. https://doi.org/10.1186/1471-2180-11-244 PMID: 22047020
- Lewis DA, Brown R, Williams J, White W, Jacobson SK, Marchesi JR, et al. The human urinary microbiome; bacterial DNA in voided urine of asymptomatic adults. Frontiers in Cellular and Infection Microbiology 2013 Aug 15 https://doi.org/10.3389/fcimb.2013.00041 PMID: 23967406
- Whiteside SA, Razvi H, Dave S, Reid G, Burton JP. The microbiome of the urinary tract—a role beyond infection Nat. Rev. Urol. 2015; 12: 81–90. https://doi.org/10.1038/nrurol.2014.361 PMID: 25600098
- 22. Human Microbiome Project Consortium. Structure, function and diversity of the healthy human microbiome. Nature 2012; 486: 207–214. https://doi.org/10.1038/nature11234 PMID: 22699609
- 23. Morgan XC, Huttenhower C. Human microbiome analysis. PLoS Computational Biology 2012; 8(12): e1002808. https://doi.org/10.1371/journal.pcbi.1002808 PMID: 23300406
- Bhute S, Pande P, Shetty SA, Shelar R, Mane S, Kumbhare SV, et al. Molecular characterization and meta-analysis of gut microbial communities illustrate enrichment of *Prevotella* and *Megasphaera* in Indian subjects. Frontiers in Microbiology 2016 May 9. https://doi.org/10.3389/fmicb.2016.00660 PMID: 27242691
- Hamburger J, He-Na Abbasi I, Ramzy RM, Jourdane J, Ruppel A. Polymerase chain reaction assay based on a highly repeated sequence of *Schistosoma haematobium*: A potential tool for monitoring schistosome-infested water. Am. J. Trop. Med. Hyg. 2002; 65(6): 907–11.
- 26. Richter J, Hatz C, Campagne G, Bergquist N, Jenkins J. Ultrasound in schistosomiasis, A practical guide to the standardized use of ultrasonography for assessment of schistosomiasis-related morbidity. 2000 May. Available from www.who.int/schistosomiasis/resources/tdr_str_sch_00.1/en/
- Caporaso JG, Kuczynski J, Stombaugh J, Bittinger K, Bushman FD, Costello EK, et al. QIIME allows analysis of high-throughput community sequencing data. Nature Methods 2010; 7: 335–336. https:// doi.org/10.1038/nmeth.f.303 PMID: 20383131
- DeSantis TZ, Hugenholtz P, Larsen N, Rojas M, Brodie EL, Keller K, et al.Greengenes, a chimerachecked 16S rRNA gene database and workbench compatible with ARB. Applied and Environmental Microbiology 2006; 72: 5069–5072 https://doi.org/10.1128/AEM.03006-05 PMID: 16820507
- Salter SJ, Cox MJ, Turek EM, Calus ST, Cookson WO, Moffatt MF, et al. Reagent and laboratory contamination can critically impact sequence-based microbiome analyses. BMC Biology 2014; 12: 87–96. https://doi.org/10.1186/s12915-014-0087-z PMID: 25387460
- Jervis-Brady J, Leong LEX, Marri S, Smith RJ, Choo JM, Smith-Vaughan HC, et al., Deriving accurate microbiota profiles from human samples with low bacterial content through post-sequencing processing of Illumina MiSeq data. Microbiome 2015; 3: 19. <u>https://doi.org/10.1186/s40168-015-0083-8</u> PMID: 25969736
- Parks DH, Tyson GW, Hugenholtz P, Beiko RG. STAMP: statistical analysis of taxonomic and functional profiles. Bioinformatics 2014 Jul 23. https://doi.org/10.1093/bioinformatics/btu494 PMID: 25061070
- Oksanen J, Blanchet FG, Friendly M, Roeland K, legendre P, McGlinn D, et al. Vegan: Community Ecology Package. R package version 2.4–0. 2016; Sep 7. Available from https://CRAN.R-project.org/package=vegan
- **33.** McMurdie PJ, Paulson J. Biom: An interface package for the BIOM fileformat. R package version 0.4.0. 2014; Feb 24. Available from http://biom-format.org/
- Seagata N, Izard J, Waldron L, Gevers D, Miropolsky L, Garrett WS, et al. Metagenomic biomarker discovery and explanation. Genome Biology 2011; 12:R60. <u>https://doi.org/10.1186/gb-2011-12-6-r60</u> PMID: 21702898
- White JR, Nagarajan N, Pop M. Statistical methods for detecting differentially abundant features in clinical metagenomic samples. PLoSComput Biol. 2009; 5:e1000352. https://doi.org/10.1371/journal.pcbi. 1000352 PMID: 19360128
- McMurdie PJ, Holmes S. Waste not, want not: Why rarefying microbiome data is inadmissible. PLoS Computational Biology 2014; 10(4):e1003531. https://doi.org/10.1371/journal.pcbi.1003531 PMID: 24699258

- Robinson MD, McCarthy DJ, Smyth GK. edgeR: a Bioconductor package for differential expression analysis of digital gene expression data. Bioinformatics 2009; 26: 139–140. https://doi.org/10.1093/ bioinformatics/btp616 PMID: 19910308
- Langille MGI, Zaneveld J, Caporaso JG, McDonald D, Knights D, Reyes JA, et al. Predictive functional profiling of microbial communities using 16S rRNA marker gene sequences. Nature Biotechnology 2013; 9: 814–822.
- Pearce MM, Hilt EE, Rosenfeld AB, Zilliox MJ, Thomas-White K, Fok C, et al. The female urinary microbiome: a comparison of women with and without urgency urinary incontinence. mBio 2014; 5(4): e01283–14. https://doi.org/10.1128/mBio.01283-14 PMID: 25006228
- 40. Willner D, Low S, Steen JA, George N, Nimmo GR, Schembri MA, et al. Single clinical isolates from acute uncomplicated urinary tract infections are representative of dominant in situ populations. mBio 2014; 5(2):e01064–13. https://doi.org/10.1128/mBio.01064-13 PMID: 24570371
- Nelson DE, Van Der Pol B, Dong Q, Revanna KV, Fab B, Easwaran S, et al. Characteristic male urine microbiomes associate with asymptomatic sexually transmitted infection. PLoS ONE 2010; 5(11): e14116. https://doi.org/10.1371/journal.pone.0014116 PMID: 21124791
- Adeyeba OA, Ojeaga SGT. Urinary schistosomiasis and concomitant urinary tract pathogens among school children in metropolitan Ibadan, Nigeria. Afr. J. Biomed. Res. 2002; 5: 103–107.
- Siddiqui H, Lagesen K, Nederbragt AJ, Jeansson SL, JakobsenKS. Alterations of microbiota in urine from women with interstitial cystitis.BMC Microbiology 2012; 12: 205–16. https://doi.org/10.1186/1471-2180-12-205 PMID: 22974186
- Ott SJ, Musfeldt M, Wenderoth DF, Hampe J, Brant O, Folsch UR, et al. Reduction in diversity of the colonic mucosa associated bacterial microflora in patients with active inflammatory bowel disease. Gut 2004; 53(5): 685–693. https://doi.org/10.1136/gut.2003.025403 PMID: 15082587
- 45. Fouts DE, Pieper R, Szpakowski S, Pohl H, Knolbach S, Suh MJ, et al. Integrated next-generation sequencing of 16S rDNA and metaproteomics differentiate the healthy urine microbiome from asymptomatic bacteriuria in neuropathic bladder associated with spinal cord injury. Journal of Translational Medicine 2012; 10: 174–190. https://doi.org/10.1186/1479-5876-10-174 PMID: 22929533
- Kahlmeter G. An international survey of the antimicrobial susceptibility of pathogens from uncomplicated urinary tract infections: the ECO• SENS Project. Journal of Antimicrobial Chemotherapy 2003; 51: 69–76. PMID: 12493789
- Bergstrand LH, Cardenas E, Holert J, Van Hamme JD, Mohn WW. Delineation of steroid degrading microorganisms through comparative genomic analysis. mBio 2016; 7(2):e00166–16. https://doi.org/ 10.1128/mBio.00166-16 PMID: 26956583
- Tian S, Ali M, Xie L, Li L. Complete genome sequence analysis of Acinetobacterjohnsonii reveals potential nematode-virulent factors. Springerplus 2016 Jul 4. https://doi.org/10.1186/s40064-016-2668-5
- Kostic AD, Chun E, Robertson L, Glickman JN, Gallini CA, Michaud M, et al. *Fusobacterium* potentiates intestinal tumorigenesis and modulates the tumour-immune microenvironment. Cell Host & Microbe 2013; 14: 207–215.
- Kostic AD, Gevers D, Pedamallu CS, Michaud M, Duke F, Earl AM, et al. Genomic analysis identifies association of *Fusobacterium* with colorectal carcinoma. Genome Research 2012; 22: 292–298. https:// doi.org/10.1101/gr.126573.111 PMID: 22009990
- Fujiwara N, Porcelli SA, Naka T, Yano I, Maeda S, Kuwata H, et al. Bacterial sphingophospholipids containing non-hydroxy fatty acid activate murine macrophages via Toll-like receptor 4 and stimulate bacterial clearance. BiochimicaetBiophysicaActa 2013; 1831: 1177–1184.
- Choi EJ, Iwasa M, Han KI, Kim WJ, Tang Y, Hwang YJ, et al. Effect of *Enterococcusfaecalis* EF-2001 on experimentally induced atopic eczema in mice. Food Sci. Biotechnol 2016; 25: 1087. https://doi.org/ 10.1007/s10068-016-0175-7
- Shimada T, Cheng L, Yamasaki A, Ide M, Motonaga C, Yasueda H, et al. Effects of lysed *Enterococcus*faecalis on allergen-induced serum antibody responses and active cutaneous anaphylaxis in mice. Clin. Exp. Allergy 2004; 34(11): 1784–8. <u>https://doi.org/10.1111/j.1365-2222.2004.02092.x</u> PMID: 15544605
- Arinola OG. Immunological Aspects of Schistosomiasis. Annals of Ibadan Postgraduate Medicine 2005; 3(1): 69–73.
- Hakansson A, Molin G. Gut Microbiota and Inflammation. Nutrients 2011; 3(6): 637–682. <u>https://doi.org/10.3390/nu3060637</u> PMID: 22254115
- 56. Troy EB, Kasper DL. Beneficial effects of *Bacteroides* fragilis polysaccharides on the immune system. Front Biosci. 2010; 15: 25–34.

- Fahmy N, Lazo-Langner A, lansavichene AE, Pautler SE. Effect of anticoagulants and antiplatelet agents on the efficacy of intravesical BCG treatment of bladder cancer: a systematic review. Can. Urol. Assoc. J. 2013; 7: E740–E749. https://doi.org/10.5489/cuaj.1213 PMID: 24282468
- Seow SW, Rahmat JN, Mohamed AA, Mahendran R, Lee YK, Bay BH. Lactobacillus species is more cytotoxic to human bladder cancer cells than Mycobacterium bovis (bacillus Calmette-Guérin). J. Urol. 2002; 168: 2236–2239. https://doi.org/10.1097/01.ju.0000034353.97729.69 PMID: 12394766
- 59. Maskell RM. The natural history of urinary tract infection in women. Med. Hypotheses 2010; 74(5): 802–806. https://doi.org/10.1016/j.mehy.2009.12.011 PMID: 20064694
- Winter SE, Lopez CA, Baumler AJ. The dynamics of gut-associated microbial communities during inflammation. EMBO Rep 2013; 14: 319–327. https://doi.org/10.1038/embor.2013.27 PMID: 23478337
- Lee KW, Park JY, Jeong HR, Heo HJ, Han NS, Kim JH. Probiotic properties of *Weissella* strains isolated from human faeces. Anaerobe. 2012; 18(1): 96–102. https://doi.org/10.1016/j.anaerobe.2011.12.015 PMID: 22200451
- Sapountzis P, Gruntjes T, Otani S, Estevez J, da Costa RR, Plunkett G, et al. The Enterobacterium *Trabulsiella* odontotermitis present novel adaptations related to its association with fungus-growing termites. Appl Environ Microbiol. 2015; 81(19): 6577–88. https://doi.org/10.1128/AEM.01844-15 PMID: 26162887
- Masuelli L, Pantanella F, La Regina G, Benvenuto M, Fantini M, Mattera R, et al. Violacein, an indolederived purple-colored natural pigment produced by *Janthinobacteriumlividum*, inhibits the growth of head and neck carcinoma cell lines both in vitro and in vivo. Tumor Biology 2016; 37: 3705–3717. https://doi.org/10.1007/s13277-015-4207-3 PMID: 26462840
- **64.** Forst S, Dowds B, BoemareN. *Xenorhabdus* and Photorhabdus spp.: bugs that kill bugs. Annual Reviews in Microbiology 1997; 51: 47–72.
- Makarova K, Slesarev A, Wolf Y, Sorokin A, Mirkin B, Koonin E, et al. Comparative genomics of the lactic acid bacteria. Proceedings of the National Academy of Sciences 2006; 103: 15611–15616.
- Sicard M, Brugirard-Ricaud K, Pagès S, Lanois A, Boemare NE, Brehelin M, et al. Stages of infection during the tripartite interaction between *Xenorhabdus* nematophila, its nematode vector, and insect hosts. Applied and Environmental Microbiology 2004; 70: 6473–6480. <u>https://doi.org/10.1128/AEM.70. 11.6473-6480.2004 PMID: 15528508</u>
- **67.** Börnigen D, Morgan XC, Franzosa EA, Ren B, Xavier RJ, Garrett WS, et al. Functional profiling of the gut microbiome in disease-associated inflammation. Genome Medicine 2013; 5:65 https://doi.org/10. 1186/gm469 PMID: 23906180
- Ziebart KT, Dixon SM, Avila B, El-Badri MH, Guggenheim KG, Kurth MJ, et al. Targeting multiple chorismate-utilizing enzymes with a single inhibitor: validation of a three-stage design. Journal of Medicinal Chemistry 2010; 53: 3718–3729. https://doi.org/10.1021/jm100158v PMID: 20359225
- 69. Behnsen J, Raffatellu M.Siderophores: More than Stealing Iron. MBio 2016; 7: e01906–01916.
- Holden VI, Bachman MA. Diverging roles of bacterial siderophores during infection. Metallomics2015; 7: 986–995. https://doi.org/10.1039/c4mt00333k PMID: 25745886
- Wolfe AJ, Toh E, Shibata N, Rong R, Kenton K, Fitzgearld M, et al. Evidence of uncultivated bacteria in the adult female bladder. ClinMicrobiol. 2012; 50(4): 1376–83.